Author's response to reviews

Title: Tuberculosis screening and follow-up of asylum seekers in Norway. A cohort study.

Authors:

Ingunn IH Harstad (ingunn.harstad@ntnu.no)
Einar EH Heldal (einar.heldal@c2i.net)
Sigurd SLS Steinshamn (sigurd.steinshamn@ntnu.no)
Helge HG Garåsen (helge.garasen@ntnu.no)
Geir GWJ Jacobsen (geir.jacobsen@ntnu.no)

Version: 2 Date: 16 October 2008

Author's response to reviews:

There was an error in the numbers at reference number 8 that is corrected in this version